Author's response to reviews

Title: Involvement of TSC genes and differential expression of other members of the insulin signaling pathway in oral squamous cell carcinoma

Authors:

Sanjukta Chakraborty (sanjuc8@gmail.com)
SM Azeem Mohiyuddin (azeem_hn@hayoo.co.in)
K S Gopinath (dr.gopinath@hotmail.com)
Arun Kumar (karun@mrdg.iisc.ernet.in)

Version: 4 Date: 16 May 2008

Author's response to reviews:

Dr. Ashleigh Manning
Assistant Editor
BMC-series journals

Dear Dr. Manning,

I am sending our manuscript entitled “Involvement of TSC genes and differential expression of other members of the mTOR signaling pathway in oral squamous cell carcinoma” for publication in “BMC Cancer”. The re-revised manuscript has five figures and two tables. I have inserted Table 1 after the References section in the main body of the manuscript as suggested. However, Table 2 has been provided as a separate file as it is large. The manuscript has been re-revised according to the suggestion of the Reviewer#1. We have also formatted the manuscript according to the journal style.

Reviewer 1 (Dr. Eloiza Helena Tajara):

Comment: I have only one comment.
Page 15, last sentence: The data from the study of Hebert et al. should be presented at the beginning of page 13.

Our response: We have incorporated this suggestion in the manuscript.

Thanks for your interest in our manuscript and accepting it for publication in "BMC Cancer".

I would be happy to pay the processing charges. Should you have any question, please let me know.

Warm regards,

Sincerely,
Prof. Arun Kumar, Ph.D.
Associate Professor